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/FD: Detector Assembly Progress =
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@  The Intensity Frontier NOVA Far Detector Assembly Progress

Status Date: 28PR14

100% Filled
April, 25, 2014
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FarDet: Data-Taking >
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DB 09 & 10 about to go to Partitionl this weak

>
Partition 2 — Diblocks 9,10,11* +1 Partition 1 — diblocks 2,3,4,5,6,7,8
Diblocks 1, 9,10 cooled APDs Running ~99% active, cooled APDs
Installation of new APDs on DB 11 HYV at full gain values

Partition 3 — running diblocks 12,13 (Front End Boards only)
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/FD: DAQ Uptime !
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» Plot for past 4 weeks, much better than last week.

> Averaging about 86% and good ~ 91% in past week.

» Mon: rearrangement of diblocks for physics partition + pedestal runs.
» Thu: maintenance day, repairs work on APDs/FEBs.

» Fri: Setting of full gain HV mode and cooling down swapped parts.
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/FD: POTs delivered/recorded
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> Plot for past 4 weeks, efficiency went up.

» Otherwise correlated to DAQ downtimes (Thu & Fri).
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> We recorded in average 0.93 e18 (from 1.03) pots/day with 91% efficiency.
» Over weekend (When no installation/ reapairs) we achieved 100%.
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/Summary A

AR A
FarDet:
® Good news: 100% of Far detector filled with o1l (Apr 25).

* 20.7 of 28 blocks of FarDet fully-instrumented, running with
full gain mode, more than 27.8 blocks oil filled.

e 10 of 14 diblocks successfully cooled down.

e The ‘Physics’ partition was enlarged to 8 diblocks last week.

* Expected this week DB 09&10, DB 01 went away for
replacement of uncalibrated FEBs.

» This completed three “Super-block” .
> We will running with 3 superblocks (2—4) for neutrino hunting.

* Efficiency of beam recording correlated with DAQ down times,
in general we had better week, no major issues.

NearDet:

® Good news: 100% of Near detector filled with oil (Apr 24).

¢ Installed first 8 APDs onto p-catcher to test full DAQ chain.

* >350 FEB5 are installed on ND (>50%), 300 on the way (110%)
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